Author’s response to reviews

Title: Macular Hole Retinal Detachment After Intravitreal Conbercept Injection for the Treatment of Choroidal Neovascularization Secondary to Degenerative Myopia: a Case Report

Authors:
Chuanbin Sun (sunshine2012@zju.edu.cn)
Yueye Wang (wangyy@126.com)
shiyang Zhou (zhousy@126.com)
Xudong Fang (fangxd@126.com)
Danni Xu (xudn@126.com)
Zhe Liu (doctorliuzhe@126.com)

Version: 3 Date: 30 Jun 2019

Author’s response to reviews:

Dear Editor:

Thank you very much for your valuable comments on our manuscript entitled “macular hole retinal detachment after intravitreal conbercept injection for the treatment of choroidal neovascularization secondary to degenerative myopia: a case report” (BOPH-D-19-00332). According to your comments, we revised our manuscript again, all modifications in the revised manuscript were highlighted with red letters. The responses to your comments are listed as follows:

Comments: The manuscript reports a rare clinical presentation which is worthy of reporting; however, there are several English language corrections which need to be made and the authors need to have it proofread to correct all such issues. A word processor English grammar check also needs to be performed.

For instance, in the first sentence in the abstract "firstly" should be removed.

In the Background, the first sentence should read "Choroidal neovascularization is characteristic 

Response: we appreciate Editor’s valuable comments. Based on your comments, we checked our manuscript again and found some grammar and typing errors, which were corrected in the revised manuscript highlighted with red letters. All corrections are listed as follows:
1. "firstly" in “Abstract Section, Line 1, Page 2” in the original manuscript is deleted in the revised manuscript;

2. “Uneventful intravitreal …… in the left eye” in “Abstract Section, Line 6-8, Page 2” in the original manuscript is modified to be “Intravitreal injection …… in the left eye” in “Abstract Section, Line 6-8, Page 2” in the revised manuscript;

3. “unfortunately a full thickness MH …… by spectral-domain optical coherence tomography” in “Abstract Section, Line 8-9, Page 2” in the original manuscript is modified to be “Unfortunately, a full thickness MH …… by ophthalmoscopy and spectral-domain optical coherence tomography” in “Abstract Section, Line 8-10, Page 2” in the revised manuscript;

4. “Vitrectomy and silicone oil tamponade …… during three months’ follow-up” in “Abstract Section, Line 9-11, Page 2” in the original manuscript is modified to be “Vitrectomy and silicone oil tamponade …… during three months’ follow-up” in “Abstract Section, Line 10-12, Page 2” in the revised manuscript;

5. "Choroidal neovascularization (CNV) is characteristics of …… " in “Background Section, Line 2, Page 3” in the original manuscript is modified to be "Choroidal neovascularization (CNV) is characteristic of ……” in “Background Section, Line 2, Page 3” in the revised manuscript;

6. “which were mostly reported in anti-VEGF therapy……”in “Background Section, Line 8, Page 3” in the original manuscript is modified to be "which was mostly reported in anti-VEGF therapy ……” in “Background Section, Line 8, Page 3” in the revised manuscript;

7. “MH was still open accompanied by macular retina reattachment and Fuchs spot formation……” in “Background Section, Line 14-15, Page 3” in the original manuscript is modified to be “MH still kept open despite successful macular reattachment and Fuchs spot formation……” in “Background Section, Line 14-15, Page 3” in the revised manuscript;

8. “color fundus photograph demonstrated that MH was still open accompanied by macular retina reattachment” in “Figure Legend Section, Line 19-20, Page 10” in the original manuscript is modified to be “color fundus photograph demonstrated that MH still kept open despite successful macular reattachment” in “Figure Legend Section, Line 19-20, Page 10” in the revised manuscript.

We thank you again for your kind consideration of our manuscript for publication in your prestigious journal. Please do not hesitate to contact me if there are any further questions or concerns.

Best Regards.
Liu Zhe. MD

Department of Ophthalmology, Zhejiang Provincial People’s Hospital, Hangzhou 310015, China.
E-mail: doctorliuzhe@126.com